

## OPEN ACCESS

**Citation:** Sánchez MI, Pons I, Martínez-Haro M, Taggart MA, Lenormand T, Green AJ (2016) When Parasites Are Good for Health: Cestode Parasitism Increases Resistance to Arsenic in Brine Shrimps. PLoS Pathog 12(3): e1005459. doi:10.1371/journal. ppat.1005459

**Editor:** Catarina Eira, Universidade de Aveiro, PORTUGAL

Received: October 20, 2015

Accepted: January 25, 2016

Published: March 3, 2016

**Copyright:** © 2016 Sánchez et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper or can be found at <u>http://digital.csic.</u>es/handle/10261/128374.

**Funding:** This work was funded by the Spanish Ministry of Economy and Competitiveness through the Severo Ochoa Program for Centres of Excellence in R+D+I (SEV-2012-0262) and the I+D+i National Plan (Project CGL2013-47674-P). MIS was supported by a Ramón y Cajal postdoctoral contract from the Spanish Ministry of Science and Innovation (MICINN). MMH was supported by a Marie Curie Intra-European Fellowship for Career Development **RESEARCH ARTICLE** 

# When Parasites Are Good for Health: Cestode Parasitism Increases Resistance to Arsenic in Brine Shrimps

Marta I. Sánchez<sup>1®</sup>\*, Inès Pons<sup>1®</sup>, Mónica Martínez-Haro<sup>2,3</sup>, Mark A. Taggart<sup>4</sup>, Thomas Lenormand<sup>5</sup>, Andy J. Green<sup>1</sup>

 Department of Wetland Ecology, Estación Biológica de Doñana, (EBD-CSIC), Seville, Spain,
 Department of Life Sciences, Marine and Environmental Sciences Centre (MARE), University of Coimbra, Coimbra, Portugal, 3 Instituto de Investigación en Recursos Cinegéticos (IREC-CSIC-UCLM-JCCM), Ciudad Real, Spain, 4 Environmental Contamination and Ecological Health, Environmental Research Institute, University of the Highlands and Islands, Thurso, Scotland, United Kingdom, 5 Department of Genetic and Evolutive Ecology, Center of Functional Ecology and Evolution (CEFE), National Centre for Scientific Research (CNRS), Montpellier, France

These authors contributed equally to this work.
 \* marta.sanchez@ebd.csic.es

## Abstract

Parasites and pollutants can both affect any living organism, and their interactions can be very important. To date, repeated studies have found that parasites and heavy metals or metalloids both have important negative effects on the health of animals, often in a synergistic manner. Here, we show for the first time that parasites can increase host resistance to metalloid arsenic, focusing on a clonal population of brine shrimp from the contaminated Odiel and Tinto estuary in SW Spain. We studied the effect of cestodes on the response of Artemia to arsenic (acute toxicity tests, 24h LC<sub>50</sub>) and found that infection consistently reduced mortality across a range of arsenic concentrations. An increase from 25°C to 29°C, simulating the change in mean temperature expected under climate change, increased arsenic toxicity, but the benefits of infection persisted. Infected individuals showed higher levels of catalase and glutathione reductase activity, antioxidant enzymes with a very important role in the protection against oxidative stress. Levels of TBARS were unaffected by parasites, suggesting that infection is not associated with oxidative damage. Moreover, infected Artemia had a higher number of carotenoid-rich lipid droplets which may also protect the host through the "survival of the fattest" principle and the antioxidant potential of carotenoids. This study illustrates the need to consider the multi-stress context (contaminants and temperature increase) in which host-parasite interactions occur.

## Author Summary

Virtually all free-living organisms are infected by parasites. Moreover, both parasites and hosts may be exposed to increasing levels of pollution and might be affected by climate change. However, few studies have considered the environmental context in which

(PIEF-GA-2011-299747) within the 7th Framework Programme (FP7 2007-2013) of the European Commission. Currently, MMH benefits from a postdoctoral contract funded by the Junta de Comunidades de Castilla-La Mancha (POST 2014/ 7780). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

parasites and hosts interact, and the relationships between these factors remains poorly understood. It is assumed that infection with parasites increases mortality under a cause of stress such as pollution. We studied the combined effect of arsenic (As) pollution, temperature increase and infection by tapeworms on the health of the economically and ecologically important brine shrimp *Artemia*. We found that tapeworms make *Artemia more* resistant to As, a major pollutant in aquatic environments, even under increased temperature conditions. These parasites increase the capacity of antioxidant enzymatic defenses, allowing infected individuals to cope better with As. Moreover, tapeworms increase fat reserves in their hosts, which may be advantageous due to the ability of lipids to sequester pollutants ("survival of the fattest" principle). Although our results may be unusual, we find a clear explanation for them. This makes them of broad significance and general interest.

## Introduction

Many aspects of host-parasite interactions have been studied in detail, from molecular mechanisms to adaptive strategies and their ecological and evolutionary consequences (reviewed in Schmid-Hempel 2011 [1]). In contrast, few studies have considered the context of multiple environmental stressors in which host-parasite interactions occur in natural conditions. Consequently there are important limitations to our understanding of the ecology and evolution of host-parasite interactions, and to our ability to reach reliable conclusions. For example, increasing numbers of taxa (both, free living and parasites) are exposed to pollution and impacted by climate change [2]. However, the complex relationships between these factors (i.e. parasites, pollution and climate change) are not well understood.

Parasites and hosts can react differently to pollutants, influencing their mutual interactions. For instance, if the parasite is more susceptible than the host to the pollutant, the exposure to pollution provides the indirect benefit (for the host) of protecting against the parasite. Conversely, the epidemiology of the parasite may be altered negatively if the pollutant impacts the life history of the host (e.g. reducing survival), thus compromising parasite transmission. Moreover, parasites and pollution can interact to affect the health of the host, the central topic of the emerging field of "Environmental Parasitology" [3]. Most studies evaluating the joint effect of parasites and pollution on the health of free-living organisms find that there are additive or synergistic effects between these stressors [4,5,6]. For example, coinfection of amphipods by acanthocephalans and microsporidians led to a reduction in antitoxic defenses when exposed to cadmium [7].

However, parasites can influence multiple facets of host phenotype [8], including physiology, behavior and biochemistry, so they may change the host response to a pollutant in diverse and complex ways. The physiology of the parasite itself may have a direct influence, e.g. through the capacity observed in several parasite groups to bioaccumulate contaminants [9,10]. Therefore, more studies are required to understand the diversity of ways parasites and pollution can interact to affect the health of organisms. Of particular concern are the mechanistic (proximate) bases of host-parasite interactions; for example the potential of oxygen-free radicals and other reactive oxygen species (ROS) to induce oxidative damage in tissues and cellular components, leading to adverse health effects and diseases. The evaluation of the levels of antioxidant enzymes in infected and uninfected organisms can provide valuable information on antioxidant status and their capacity to confront multiple stress conditions. Furthermore, all these effects are likely to depend on physiological variation which is influenced by temperature and hence by climate change. The projected temperature increase this century [11] is likely to increase the toxicity of pollutants [12] as well as the transmission, distribution and abundance of many parasites [13,14,15]. To date, studies of interactions between parasites and pollutants on hosts have only been carried out at a single temperature, with the exception of one investigation [16] dealing with the effect of trematode infection and seasonal temperature on bioaccumulation of xenobiotics in freshwater clams.

Brine shrimps *Artemia* spp. (Branchiopoda, Anostraca) are economically and ecologically important. They are used as model organisms in aquatic toxicology [<u>17</u>, <u>18</u>] and are the most important live food used in aquaculture worldwide [<u>19</u>]. *Artemia* are the dominant macroinvertebrate in hypersaline ecosystems around the world, and many waterbirds depend on them as food [<u>20,21,22</u>]. They control phytoplankton populations [<u>23</u>] and are also intermediate hosts for a rich community of avian cestodes [<u>24</u>]. These parasites cause strong physiological and behavioural changes in *Artemia* [<u>25,26,27</u>] which may be expected to influence their response to pollutants.

We studied the *Artemia parthenogenetica* population from the Odiel and Tinto estuary, SW Spain, one of the most polluted estuarine systems in Western Europe [28]. *A. parthenogenetica* from Odiel have high levels of cestode infection [29], especially of *Flamingolepis liguloides* which uses flamingos as final hosts, and *Confluaria podicipina* which infects grebes [30]. Arsenic is a highly toxic and bioaccumulable metalloid that originates from both anthropogenic and natural sources and causes detrimental effects in humans and wildlife [31]. Arsenic in the Odiel and Tinto estuary originates from historic and current mining activity [32,33,34]. An estimated 12 t yr<sup>-1</sup> and 23 t yr<sup>-1</sup> of As is transported by the Tinto and Odiel rivers (respectively) into the Atlantic Ocean [35]. Previous studies have found high levels of inorganic arsenic in sediments (85–610 ppm) in the study area [36], exceeding the ERM (Effects Range Median) for marine and estuarine sediments (70 ppm [37]), and also the Canadian sediment quality guide-line (CSQG) value for the protection of aquatic life (7.2 ppm [38]).

This study was designed to test the individual and combined effects of infection by cestode parasites and a 4°C temperature change on the sensitivity of Artemia to acute As exposure. We conducted toxicity tests to compare As median lethal concentrations at 24 h (24-h  $LC_{50}$ ) for infected and uninfected individuals on two separate dates one month apart, in order to test the effect of different parasite assemblages on arsenic sensitivity. Since cestode parasites in A. parthenogenetica have a strong seasonal pattern [29], and different parasite species induce different physiological effects in their host, we chose to consider the effects of different parasite compositions (i.e dates). We also evaluated the effect of parasites on the antioxidant defense mechanisms of Artemia, in order to measure the capacity of infected animals for detoxification of reactive oxygen species caused by factors such as pollution or climate change. We included the activity of four important enzymes: superoxide dismutase (SOD) responsible for the detoxification of the highly reactive oxygen species superoxide anions; catalase (CAT) which catalyses the decomposition of hydrogen peroxide in  $0_2$  and  $H_20$ ; glutathione peroxidase (GPX) which acts as a scavenger for high concentrations of hydrogen peroxide; and glutathione reductase (GR) implicated in the reduction of glutathione disulphide to the sulphydryl form glutathione, which is a critical molecule in combating oxidative stress. In order to evaluate the efficacy of the above enzymatic mechanisms for control of reactive oxygen species, we also measured the levels of thiobarbituric acid reactive substances (TBARS), which indicates the extent of lipid peroxidation as a consequence of oxidative damage. Finally, we quantified the carotenoid-rich lipid droplets of infected and uninfected individuals. These droplets serve as intracellular lipid storage but also play a protective role in protecting cells against oxidative stress.



Fig 1. Infected and uninfected Artemia. Infected (red) and uninfected (transparent) Artemia parthenogenetica from Odiel and Tinto estuary.

## Results

## Acute toxicity tests (median lethal concentrations, 24-h LC<sub>50</sub>)

**Infection characteristics between sampling dates.** In order to test the effect of different parasite assemblage on the response of *Artemia* to As, two experiments were conducted in different months (April and May 2014). Infected and uninfected individuals were separated on the basis of their colour ([<u>39</u>], <u>Fig 1</u>), and their parasitic status was confirmed after the experiment.

Infection indices (prevalence and abundance) for six of nine cestode species recorded in the *infected group* varied significantly between the two experiments (<u>Table 1</u>).

Infected individuals in April were dominated by *Flamingolepis liguloides* (68%) and *Confluaria podicipina* (65%), with 36% of individuals being infected by both these parasites (Fig 2).

In contrast, infected individuals in May were dominated by *F. liguloides* (90%) with no *C. podicipina*. Other minority cestodes (prevalences of 0.2–14%) present in one or both of the experiments were *Flamingolepis flamingo* and *Gynandrotaenia stammeri* (with flamingos as final hosts), *Eurycestus avoceti*, *Anomotaenia microphallos* and *A. tringae* (shorebirds), *Fimbriarioides tadornae* (shelducks) and *Wardium stellorae* (gulls). Total parasite loads were significantly higher in April, as was the number of cestode species present per individual *Artemia* (<u>Table 1</u>). The proportion of infected *Artemia* with only a single cysticercoid increased from

| Parasite species         | Infection descriptor | 14-April       | 15-May          | Z test  | M-W U test  |
|--------------------------|----------------------|----------------|-----------------|---------|-------------|
| Flamingolepis liguloides | P (%)                | 68.00 ± 2.6    | 89.8 ± 1.2      | 8.52 *  |             |
|                          | MA                   | 1.20 ± 0.065   | 1.50 ± 0.041    |         | 128423.5 ** |
| Flamingolepis flamingo   | P (%)                | 6.00 ± 1.3     | $0.90 \pm 0.4$  | 21.57 * |             |
|                          | MA                   | 0.058 ± 0.013  | 0.015 ± 0.0083  |         | 103318.5 ** |
| Confluaria podicipina    | P (%)                | 65.20 ± 2.6    | 0.00            | 23.29 * |             |
|                          | MA                   | 0.74 ± 0.036   | -               |         | 37835 **    |
| Wardium stellorae        | P (%)                | 3.00 ± 1.00    | 1.70 ± 0.5      | 13.22 * |             |
|                          | MA                   | 0.033 ± 0.0099 | 0.017 ± 0.005   |         | 106766      |
| Anomotaenia microphallos | P (%)                | $1.00 \pm 0.5$ | 7.00 ± 1.00     | 1.52    |             |
|                          | MA                   | 0.0091 ±0.0052 | 0.071 ± 0.011   |         | 115011 **   |
| Anomotaenia tringae      | P (%)                | 2.1 ± 0.8      | 12.30 ± 1.3     | 5.15 *  |             |
|                          | MA                   | 0.024 ± 0.0085 | 0.12 ± 0.013    |         | 119138 **   |
| Eurycestus avoceti       | P (%)                | 13.3 ± 1.9     | $5.20 \pm 0.9$  | 4.19 *  |             |
|                          | MA                   | 0.13 ± 0.019   | 0.49 ± 0.0084   |         | 99374 **    |
| Fimbriarioides tadornae  | P (%)                | 0              | $0.20 \pm 0.2$  | -0.11   |             |
|                          | MA                   | -              | 0.0015 ± 0.0015 |         | 108735      |
| Gynandrotaenia stammeri  | P (%)                | 0              | $0.20 \pm 0.2$  | -0.11   |             |
|                          | MA                   | -              | 0.0015 ± 0.0015 |         | 108735      |
| TOTAL PARASITES          | P (%)                | 98.00 ± 0.7    | 98.00 ± 0.5     | -0.81   |             |
|                          | MA                   | 2.2 ± 0.012    | 1.78 ± 0.0079   |         | 8437739**   |
| Species Richness         |                      | 1.61 ± 0.03    | 1.24 ± 0.02     |         | 71953**     |

### Table 1. Infection indexes of Artemia at different dates in 2014.

Prevalence (P%), Mean Abundance (MA ± SE) and Species Richness (± SE) of cestodes in *Artemia* samples assigned to infected experimental treatments based on coloration. A Z test was used to compare Prevalence between dates and a Mann-Whitney U test to compare mean abundance and species richness between dates.

\* P < 0.01

\*\* P < 0.001

doi:10.1371/journal.ppat.1005459.t001

35.9% in April to 52.2% in May. Likewise, the proportion of infected *Artemia* infected by only a single cestode species increased from 46.3% in April to 77.1% in May.

Median lethal concentration—LC<sub>50</sub>. We conducted acute toxicity tests (median lethal concentration, 24-h LC<sub>50</sub>) in infected and uninfected individuals at a single temperature (25°C in April) or at two temperature conditions (25 and 29°C in May). In April, when *Artemia* starts the first generation of the year, there were not enough individuals to conduct the experiment at 29°C. However it was a good moment to conduct the experiments because at the beginning of the season a high proportion of *Artemia* are infected by only one parasite species, making results easier to interpret (later in the year, mixed infections are rather common, with of up to 8 cestode species coexisting in a single host [29]). Mortality of infected *A. parthenogenetica* was consistently lower than that of uninfected *Artemia* under different As concentrations (Fig 3).

 $LC_{50}$  was significantly higher in infected individuals for both dates and parasite assemblages, and at both temperatures, as indicated by clear separation of confidence intervals [40] (Fig 4).

However, whilst the 24h LC<sub>50</sub> of the uninfected individuals at 25°C did not change between the two dates, that for infected individuals was significantly higher in May (when *F. liguloides* infection was more dominant and total parasite loads were lower) compared with the experiment conducted in April (when both *F. liguloides* and *C. podicipina* were highly prevalent) (Fig  $\underline{4}$ ). An increase in temperature decreased As tolerance in both infected and uninfected Artemia (Fig  $\underline{4}$ ).





**Fig 2.** Lipid droplets with carotenoid pigments, and hemoglobins in *Artemia* infected by *Flamingolepis liguloides* and/or *Confluaria podicipina*. *Artemia parthenogenetica* infected by cysticercoids of *Confluaria podicipina* (indicated by arrows) and *Flamingolepis liguloides* in mixed infection (A), simple infection by *C. podicipina* (B), simple infection by *F. liguloides* (C) and uninfected *Artemia* (D). Lipid droplets with carotenoid pigments are visible as orange colouration in A and B (much more pronounced in B); hemoglobins are visible as pink colouration in A and C.

In a Generalized Linear Model (GLZ), mortality rate (i.e., the number of *Artemia* that died) was significantly influenced by temperature (p = 0.015), date (p = 0.007), parasitic status (p < 0.0001) and As concentration (p < 0.0001) (<u>Table 2</u>). The interactions "parasitic status x temperature" and "As concentration x temperature" were also significant (p < 0.01 and p < 0.0001, respectively).

# Oxidative stress and lipid droplet quantification in infected and uninfected Artemia

It was impossible to analyse oxidative stress in individuals used in acute toxicity tests because of the need to crush the specimens in order to check for parasites, so we conducted a separate sampling in May 2015. Parasite prevalence has a seasonal pattern [29], so in order to have a similar parasite assemblage, samples were collected in the same calendar month as those used in toxicity tests and from the same ponds of medium-high salinity (140–220 g/l). One subsample was used for oxidative stress analysis and another to characterize parasite assemblage and to quantify lipid droplets. Infected and uninfected individuals were recognised on the basis of their colour [392]. Table 3 shows the infection index of infected individuals. As expected, infection was dominated by *F. liguloides*, and its prevalence did not differ from the sampling of May 2014 (Mann-Whitney U test, U = 13158, P = 0.585). Prevalence and abundance of minority cestode species varied among dates.

The analysis of oxidative stress in infected and uninfected *Artemia* revealed the association of parasites with strong changes in the antioxidant capacity of the host. Infected individuals



**Fig 3. Mortality of** *Artemia* **exposed to As.** Mortality (% mean  $\pm$  SE) of *Artemia* after exposure to different As concentrations (mg/l) during 24 h at 25°C (A, B) and 29°C (C), at different dates with different parasite assemblages (A: April; B, C: May). Black bars = infected individuals (n = 330, n = 330, n = 328 for A, B and C respectively); white bars = uninfected individuals (n = 328, n = 330, n = 329). Note the changes in scale on the X axis.

showed significantly higher levels of CAT (Fig 5, t = -2.892, P = 0.02) and GR activities (Fig 5, t = -2.881, P = 0.002), whereas SOD was higher in uninfected individuals (t = 2.739, P = 0.03). Conversely, parasites had a negligible effect on GPX (t = -0.814, P = 0.447) and TBARS (t = -1.781, P = 0.125).

We also quantified lipid droplets, which are readily visible throughout the body and appendices of adult *Artemia* (Fig 2B). Lipid volume estimates are strongly correlated with biochemical measurements of lipids [41], and are highly related with the ability of organisms to protect themselves against pollutants [42]. Infection with cestodes was associated with increased number of lipid droplets (Mann-Whitney U test, U = 45.5, P < 0.001, n = 20; Fig 6).



**Fig 4. Median lethal concentrations at 24 h (LC**<sub>50</sub>) of As for infected and uninfected Artemia. LC<sub>50</sub> of infected (empty squares) and uninfected (solid circles) Artemia parthenogenetica at different dates with different parasite assemblages (1 = April, dominance by *F. liguloides* and *C. podicipina*; 2 = May, dominance by *F. liguloides* with lower total infection intensity) and temperatures. Error bars correspond to 95% confidence limits.

## Discussion

## Do cestode parasites affect As toxicity in Artemia?

This study tested the effect of parasites on *Artemia* sensitivity to As, and explored the physiological crosstalk between the parasite and the host, measuring oxidative stress and lipid content in infected and uninfected *Artemia*. Our study provides the first empirical evidence that parasites can increase resistance to metal or metalloid pollution, rather than decrease it. It is also the first study to consider the influence of temperature change on parasite-pollutant interactions. In three separate acute toxicity experiments, *Artemia* infected with cestodes consistently showed lower sensitivity to As than uninfected individuals. The higher sensitivity of infected *Artemia* in April suggests that multiple infections may reduce the benefits of cestode infection to host resistance.

Our results contradict the pre-existing view that pollution and parasites are stressors that both have negative effects on the health of free living organisms. This view was based on previous field and laboratory investigations (including chronic and acute exposure to a wide variety

| Tab | ole 2. | GLZ | on | mor | tali | ty ra | ate o | of A | Artemi | ia | expo | sec | l te | 0 / | Arsen | ic. |
|-----|--------|-----|----|-----|------|-------|-------|------|--------|----|------|-----|------|-----|-------|-----|
|-----|--------|-----|----|-----|------|-------|-------|------|--------|----|------|-----|------|-----|-------|-----|

|                       | Level of Effect | Estimate  | Standard Error | Wald Stat. | р        |
|-----------------------|-----------------|-----------|----------------|------------|----------|
| Intercept             |                 | 0.368227  | 0.078794       | 21.8398    | 0.00003  |
| Temperature           | 25°C            | 0.191934  | 0.079357       | 5.8496     | 0.01558  |
| Date                  | April           | 0.086186  | 0.032191       | 7.1681     | 0.007421 |
| Infection status      | Infected        | -0.20015  | 0.031315       | 40.8507    | 0.000000 |
| As concentration      |                 | 0.022135  | 0.001391       | 253.217    | 0.000000 |
| Temp*Infection status |                 | 0.080426  | 0.031315       | 6.596      | 0.010221 |
| Temp*As concentration |                 | -0.009019 | 0.001391       | 42.0372    | 0.000000 |
| Scale parameter       |                 | 0.825806  | 0.000000       |            |          |

Generalized linear model of Artemia parthenogenetica mortality with number of dead individuals per replicate (n = 10 individuals, with three replicates per As dose) as dependent variable with a Poisson error function and log link function. Temperature 29°C, date May and uninfected status are aliased reference categories.

doi:10.1371/journal.ppat.1005459.t002

|                          | P (%) | MA            |
|--------------------------|-------|---------------|
| Flamingolepis liguloides | 81.7  | 0.92 ± 0.076  |
| Flamingolepis flamingo   | 11.7  | 0.13 ± 0.023  |
| Confluaria podicipina    | 3.3   | 0.033 ± 0.023 |
| Euricestus avoceti       | 3.3   | 0.033 ± 0.023 |

#### Table 3. Infection indices of infected Artemia in 2015.

Prevalence (P) and mean abundance (MA  $\pm$  SE) of different cestode species infecting *Artemia* parthenogenetica (n = 60) from Odiel salt ponds in May 2015.

doi:10.1371/journal.ppat.1005459.t003

of toxicants, in vertebrates and invertebrates, intermediate and definitive hosts, and in several groups of parasites [43,44,45].

The results of oxidative stress analysis provide a mechanistic explanation for our findings. Infected individuals exhibited much higher levels of CAT and GR, reflecting a superior ability to combat the effects of exposure to pollutants with oxidative potential, such as As. The particularly high levels of CAT in infected individuals (nearly double that of uninfected Artemia) is related to the increased levels of hemoglobins in F. liguloides-infected Artemia compared with uninfected individuals (Fig 2C and 2D). CAT is a haematin protein complex with four porphyrin heme groups that allow the enzyme to react with hydrogen peroxide. There is a close linear relationship between CAT activity and hemoglobin concentration in human blood [46]. CAT also has one of the highest rates of all enzymes; one CAT molecule is able to catalyse the conversion of 5 million molecules of hydrogen peroxide per second to water and oxygen. Thus, even if the level of SOD was lower in infected individuals, the control mechanisms via CAT and GR seem to be sufficient to avoid the establishment of an oxidative stress condition, as indicated by the lack of changes in TBARS between infected and uninfected individuals. TBARS is a by-product of lipid peroxidation, so this result indicates that parasites are not inducing damage by oxidative stress. Our results conflict with what most studies have shown up to now (but see for example Marcogliese et al 2005 [6]). Infection by parasites and pathogens of a wide range of taxa are generally associated with the inhibition or weakening of the host antioxidant system, and the concomitant increase in TBARS [47,48,49].

This enhancement of the anti-oxidative defense mechanisms is probably connected with the trophic transmission mode of cestodes that infect *Artemia*, which means they require "healthy" hosts in order to increase transmission success (through predation by the definitive host). Decrease in antioxidant status enhances short-term survival prospects [50], so potentiating it may be part of the transmission strategy of the parasite to increase longevity and probability of transmission (as in the "parasite manipulation hypothesis" [51]).

We also found that there were more lipid droplets in infected individuals, which is consistent with previous work indicating that *F. liguloides* increases total lipid levels [52] and that *C. podicipina* increases triglyceride levels in *Artemia* [26] (Fig 2B). Lipids have a high heavy metal binding capacity, and lipid content has a significant effect on the accumulation of As in other organisms [53]. Neutral lipids such as those in lipid droplets can protect organisms against pollutants, sequestering them away from sensitive target sites [54,55]—a principle known as *survival of the fattest* [41]. Although many studies support this principle, none have addressed parasite-mediated effects. The only previous study to suggest that parasites can increase host survival under polluted conditions through a lipid-related effect was on the freshwater clam *Pisidium amnicum* [56]. Clams infected by digenean trematode larvae are less sensitive to pentachlorophenol, perhaps because the high lipid content of the parasite changes the internal distribution of the toxicant. Pentachlorophenol is moderately lipophilic so is expected to



**Fig 5. Oxidative stress biomarkers in infected and uninfected** *Artemia.* Comparison of oxidative stress biomarkers (mean ± SE) of infected and uninfected *A. parthenogenetica* from Odiel in May 2015. Results are presented in U/mg protein (catalase, CAT; superoxide dismutase, SOD; Glutathione reductase, GR; and glutathione peroxidase, GPX) or nmol TBARS/ml. \*Statistical differences between infected and uninfected individuals with a P value < 0.05.

accumulate in adipose tissue, whereas the target sites for toxicity are the mitochondria [56]. In parasitized *Artemia*, additional lipids accumulate in the host, not in the parasite as in the case of *Pisidium*.

Lipid droplets in infected *Artemia* are associated with carotenoids (see Fig 2B) and this, together with hemoglobins induced by parasites, largely explains the colour change that allowed us to separate infected individuals with the naked eye (Fig 1). Both *F. liguloides* and *C. podicipina* increase the concentration of carotenoids in infected *Artemia* [52]. In contrast, carotenoid concentrations in other animals are often negatively correlated with parasite load [57] and with pollutants [58]. Carotenoids are potent lipid-soluble antioxidants [59] and are





doi:10.1371/journal.ppat.1005459.g006

able to inhibit lipid peroxidation in liposomes [60]. The accumulation of carotenoids in infected *Artemia* is also considered a parasite strategy to increase the probability of being predated by birds (the final host) by increasing visibility [61] and enhancing nutritive value [26]. Carotenoids provide protection against oxidative stress in many free living organisms [62,63,64] so they may also increase longevity in infected *Artemia*. Given that oxidative stress is a common marker of toxicity, not only for As in plants, invertebrates and vertebrates [65,66,67] but also for heavy metals (e.g. lead, cadmium and mercury [68]), cestode parasites may protect *Artemia* against a broader range of pollutants.

Unlike carotenoids, a positive effect of parasites on host lipid content is common in nature, e.g. in acanthocephalans infecting gammarids [69] or trematodes infecting bivalves [70]. Therefore, our finding regarding increased resistance to As in the presence of parasites may not be an isolated case, and more studies are needed to evaluate how frequently this occurs in nature.

The differences in sensitivity to As observed in infected *Artemia* collected on different dates have several possible explanations, including a negative effect of the generally higher infection levels, or a higher pathogenicity of *C. podicipina* which was absent in May. Alternatively, it could be related to seasonal changes in the ages of the parasites or their hosts, or changes in the lipid or carotenoid levels in the host. Increased heavy metal accumulation with age in cysticer-coids has been shown in other cestodes [71,72]. Previous studies of the interactions between parasites and pollutants on toxicity have focused on individual parasite species (but see Gismondi et al. [7]). In nature, co-infections of different parasites are extremely common, and our study illustrates the need to consider the effects of co-infections in environmental parasitology.

## Effect of temperature on As toxicity

The toxicity of As is highly dependent on its bioavailability, which in turn is dependent on its chemical form and the capacity to be released from environmental matrices [73]. In marine environments, As occurs predominantly as the inorganic forms arsenate (As(V)), and arsenite (As(III)), and is significantly more bioavailable from seawater and porewater than from sediments [74]. Moreover, temperature can strongly affect the chemical behaviour of pollutants and their bioavailability [75,76,77], but also the physiology of aquatic organisms. There are many studies of the effect of temperature on heavy metal toxicity, but we are not aware of any that integrate the influence of parasites. A temperature rise of 4°C caused a significant decrease in the  $LC_{50}$  for both infected and uninfected *Artemia*. The decrease in dissolved oxygen in hypersaline waters with increasing temperature coincides with higher respiratory demands in *Artemia*. Water permeability and drinking in *Artemia* increase markedly with temperature [78], hence uptake of pollutants will also increase. Indeed, copper uptake in *Artemia* increases with temperature owing to increased activity and diffusion rate [79]. Differences in As toxicity in fish have also been associated with higher uptake at higher temperatures [80].

## Conclusions

Parasites and pathogens are conventionally considered as detrimental for a host, but they can also have positive impacts with consequences for non-host species and even the whole ecosystem [81]. We provide evidence that parasites can also benefit their hosts by increasing resistance to pollutants in contaminated environments. Infection by parasites was associated with an improved antioxidant defense system (CAT and GR) without oxidative damage, as confirmed by unaffected values of TBARS. Parasites also increased the number of lipid droplets in infected individuals, which is a common phenomenon in intermediate hosts manipulated by

parasites, so more studies in other host-parasite systems are needed to evaluate the wider relevance of our findings.

Our results provide an important advance in our understanding of host-parasite interactions and underline the importance of considering interactions between parasites, pollutants and temperature in combination, particularly given expected climate change and the likelihood that toxicity will increase with temperature.

## **Material and Methods**

## Short-term toxicity of As (LC<sub>50</sub>)

Naturally infected and uninfected adults of *A. parthenogenetica* were collected with a plankton net (0.5 mm) within the Odiel saltpan complex (see Sánchez *et al.* [21] for details of the study area).

Sampling was carried out on two dates (14<sup>th</sup> of April and 15<sup>th</sup> of May 2014) at which the relative abundance of different cestode parasites changed considerably. On the 14<sup>th</sup> of April, Artemia were collected from three ponds with salinities of 140, 150 and 190 g/l (measured with a refractometer). On the 15<sup>th</sup> of May, samples were collected from another pond with a salinity of 200 g/l. These four ponds were all within the same stage of the solar evaporation process, were hydrologically interconnected, and had similar sediment type, depth, and invertebrate and bird communities [21]. These ponds were selected on the basis of the abundance of live Artemia. Artemia were transported to the laboratory in 25 litre containers. Infected and uninfected Artemia individuals of a similar size were then selected on the basis of their colour (Fig 1); infected individuals are visually recognisable by the bright red colouration induced by the cestodes, whereas uninfected individuals are practically transparent [39]. Similar size (as a proxy of age) was selected since this may influence As concentrations or sensitivity [82, 83]. There is no difference in growth rates between infected and uninfected shrimp [52]. Of the individuals visually allocated to the *infected group* prior to experiments, 98% were truly infected when inspected afterwards (n = 989). Among individuals allocated to the *uninfected* group, 98% were truly uninfected (n = 988).

Toxicity experiments were conducted after 24 h of acclimation of the *Artemia* at 100 g/l salinity (artificial salt mixture of *Instant Ocean* dissolved in distilled water). Two experiments were carried out. The first (with *Artemia* collected on 14<sup>th</sup> of April) was conducted at 25°C (the mean annual temperature in the Odiel salt pans [21]). The second (*Artemia* collected on 15<sup>th</sup> of May) was conducted at both 25 and 29°C. During summer months, *Artemia* are often exposed to temperatures  $\geq$  29°C [21, 22] and the frequency of these events is expected to increase in future.

Median lethal concentration (LC<sub>50</sub>) was used to quantify As toxicity in infected and uninfected adult *Artemia*. Arsenic, as reagent-grade sodium arsenate, NaAsO (CAS No. 10048-95-0) was used to prepare a concentrated stock solution. The study design concentrations were prepared by mixing different proportions of the stock solution and saltwater (Instant Ocean prepared with MilliQ). The saltwater used for the dilutions was prepared within 24 h of the start of the experiment and As added one hour before conducting the experiment (to prevent oxygen depletion). Ten concentrations of As between 5 and 140 mg/l were used for the experiments at 25°C (0, 5, 20, 35, 50, 65, 85, 95, 110, 125, 140 mg/l), and ten between 4 and 67 mg/l were used for the experiment at 29°C (0, 4, 11, 18, 25, 32, 39, 46, 53, 60, 67 mg/l) in order to estimate the LC<sub>50</sub>. Experimental concentrations were adjusted to the observed mortality (higher mortality at 29°C implied lower tested concentrations). These concentrations were selected after preliminary tests. Three replicates were used per concentration, with each replicate made up of a group of 10 individuals. Infected (red) and uninfected (transparent) individuals were transferred to 25 ml beakers (10 individuals per beaker) and placed in climatic chambers at the chosen temperature, with a 12:12 photoperiod and without food. After a 24 h exposure period, dead individuals (considered to be those with no movement of the appendages observed within 10 seconds) were counted and all (alive or dead) individuals were mounted on slides to confirm parasitic status under the microscope. After observations of the cysticercoids (cestode larval stage in the intermediate host) *in situ*, each infected specimen was gently pressed under the coverslip. If the identification of the cysticercoids were prepared as permanent mounts in Berlese's medium in order to facilitate observations on the morphology of rostellar hooks. Cysticercoids were identified following Georgiev *et al.* [30] and Vasileva *et al.* [84]. Prevalence (P) and mean abundance (MA) were calculated separately for the "*infected group*" on both dates.

## Oxidative stress analysis and lipid droplet quantification

*Artemia* were sampled in May 2015 from Odiel salt ponds of intermediate-high salinity. Brine shrimps were transported to the laboratory and placed in artificial sea water (Instant Ocean, 100 g/l) before the experiment. A subsample was used to characterize the exact parasite composition (n = 60 infected individuals) and quantify the number of lipid droplets (n = 20 infected and 20 uninfected *Artemia*). The numbers of cysticercoids, prevalence, mean abundance and mean intensity (see Bush et al. 1997 [85] for definitions) were calculated for each cestode species. The number of lipid droplets was estimated according to Wurtsbaugh & Gliwicz 2001 [86]. We quantified lipid levels by inspecting individuals at 30x magnification and counting the number of lipid droplets along the right side of the 5th and 6th segments of the body.

The rest of the specimens were acclimated during 24h to the experimental salinity with continuous aeration and fed *ad libitum* with lyophilized *Tetraselmis chuii* algae. The toxic concentrations of 4.69 mg/l As was selected on the basis of preliminary  $LC_{50}$  tests (the lowest concentration at which mortality was detected). Infected and uninfected *A. parthenogenetica* of the same size range were allocated to 1L glass vials (100 individuals per vial) with 600 ml of experimental solution (either control (no As) or 4.69 mg/L As) during 24h at 25°C (12:12 photoperiod) without food.

After 24h exposure, individuals were gently washed in distilled water and, after removing excess water, stored at -80°C until biochemical analysis. All operations were performed at 4°C to prevent enzyme or tissue degradation. We performed the biochemical analysis on pools of 20 individuals per treatment. Number of replicates varied from 1 to 12 depending on *Artemia* availability. The different biomarkers were determined in the whole soft tissues after homogenization and centrifugation. Tissues were homogenized with an electrical homogenator (Miccra D-1 Art Moderne Labor Technik) in cool homogenization buffer (Tris-HCl 100 mM, EDTA 0.1 mM, Triton X-100 0.1%, pH7.8) using 200 µl of buffer per sample (20 individuals). The sample was centrifuged at 14,000 rpm at 4°C for 30 minutes and supernatant stored at -80°C until enzymatic determination.

We quantified five parameters as a proxy for oxidative status of *Artemia*: activity of four enzymes (catalase CAT, superoxide dismutase SOD, glutathione peroxidase GPx and glutathione reductase GR) and lipid peroxidation levels (thiobarbituric acid reactive substances TBARS). Total protein content in the supernatant fluid was determined following a standard Bradford's procedure [87]. Enzyme activity was determined colorimetrically. Concentration of each indicator was estimated following the specific procedures below.

**Catalase (CAT) activity.** Catalase activity was analyzed according to the method of Cohen *et al.* (1970) [88] by monitoring the enzyme-catalyzed decomposition of hydrogen peroxide

using potassium permanganate. Catalase enzyme prevents the accumulation of  $H_2O_2$  by converting it to  $H_2O$  and  $O_2$  as follows:.

 $H2O2 \rightarrow 2H2O + O2$ 

Potassium permanganate (KMnO4) is a powerful oxidizing agent that decomposes hydrogen peroxide through the following reaction:

 $2\ KMnO4 + 5\ H2O2 + 3H2SO4 \rightarrow 2\ MnSO4 + 8\ H2O + 5O2 + 3K2SO4$ 

This produces a red end product. Absorbance at a wavelength of 480 nm is read five minutes after  $KMnO_4$  is added to the samples. Catalase activity is quantified by extrapolating in a standard curve of commercial catalase (SIGMA-60634), and is expressed as U/mg of total proteins.

**Superoxide dismutase (SOD) activity.** Superoxide dismutase catalyses the dismutation of superoxide radicals to oxygen and hydrogen. SOD was estimated using the xanthine-oxidase cytochrome c method as described by McCord and Fridovich (1969) [89]. The inhibition of cytochrome C was followed spectrophotometrically at 550 nm. One unit of SOD activity is defined as the amount of enzyme providing 50% inhibition of the control rate of cytochrome c reduction at 25°C. SOD activity is expressed as U/mg of total proteins.

**Glutathione peroxidase (GPx) activity.** Selenium-dependent glutathione peroxidase (Se-GPx) exerts its protective role by reducing organic peroxides and hydrogen peroxide to alcohols and water, respectively, using glutathione (GSH) as a hydrogen donor. When GSH is oxidized to GSSG it loses its protective function. The GPx activity was determined by estimating nicotinamide adenine dinucleotide phosphate (NADPH) oxidation following the method of Carmagnol et al. (1983) [90], which is a modification of the Paglia and Valentine (1967) method [91]. During the assay, the GSSG is continually reduced by an excess of glutathione reductase (GR), producing a constant level of reduced glutathione (GSH). Production of GSH from oxidised glutathione requires NADPH. Therefore, the activity of GPx is measured by the reduction in the absorbance at 340nm due to the oxidation of NADPH, and is expressed as mU/mg of total proteins.

**Glutathione reductase (GR).** Glutathione reductase (GR) plays an essential role in protection against oxidative damage by catalysing the reduction of glutathione disulphide (GSSG) in a reaction dependent on NADPH. The activity of GR was determined, as for GPx, by measuring the decrease in absorbance at 340 nm due to the oxidation of NADPH, which acts in the reduction of GSSG to GSH, following the method of Cribb et al. 1989 [92]. GR activity is expressed as mU/mg of total proteins.

**Thiobarbituric acid reactive substance (TBARS).** Lipid peroxidation was estimated by the formation of thiobarbituric acid reactive substances (TBARS) following Buege and Aust (1978) [93]. Lipid peroxidation produces malondialdehyde (MDA), which is formed from the breakdown of polyunsaturated fatty acids. MDA reacts with thiobarbituric acid, producing a red species which is optically measured at 535nm. Lower detection limit for TBARS was 0.1  $\mu$ M, and the average coefficient of variation for sample duplicates was 3.54%. TBARS was measured in mmol MDA/mg.

All biochemical analyses were performed at the Ecophysiology Laboratory of the EBD-CSIC in a Victor 3 multiplate reader (PerkinElmer, Massachusetts, USA).

### Statistical analysis

The median acute lethal concentration (LC<sub>50</sub>) and 95% confidence limits were estimated and compared between infected and uninfected individuals, different temperatures and dates using Trimmed Spearman-Karber (TSK) analysis for lethal tests [94]. The criterion of *non-overlap-ping 95% confidence limits* was used to determine a significant difference (p < 0.05) between LC<sub>50</sub> values [40].

To test the effect of date, temperature, parasitic status and As concentration on mortality we performed GLZ with a Poisson error distribution, log link function and correction for overdispersion. Date, temperature and parasitic status were included as categorical factors, and As concentration as a continuous variable. Stepwise backwards removal was used to obtain a final model containing only significant factors.

Differences in prevalence of different cestode species between the two sampling dates were evaluated with Z-tests. Comparisons of cestode abundance were performed with Mann-Whitney U tests. We also used Mann-Whitney tests to compare enzymatic activity and lipid peroxidation, as well as lipid droplets between infected and uninfected individuals. Statistica 12 software for Windows was used for all statistical analyses [95].

## Acknowledgments

E. Martínez, Director of Marismas del Odiel Natural Park, provided permission to work in the salt ponds. F. M. Miranda Castro provided technical assistance for analysis of oxidative stress in the Ecophysiology Laboratory (LEF) of the EBD-CSIC. We also thank Raquel López Luque, Isabel Carribero and Rocío Requerey for their assistance during experiments at the Aquatic Ecology Laboratory (LEA) in the EBD-CSIC.

## **Author Contributions**

Conceived and designed the experiments: MIS MMH MAT AJG TL. Performed the experiments: MIS MMH IP. Analyzed the data: MIS IP MMH AJG TL. Contributed reagents/materials/analysis tools: MIS MAT. Wrote the paper: MIS AJG.

## References

- 1. Schmid-Hempel P. Evolutionary parasitology. The integrated study of infections, immunology, ecology and genetics. Oxford. Oxford University Press. 2011.
- Brierley AS, Kingsford MJ. Impacts of Climate Change on Marine Organisms and Ecosystems. Current Biology. 2009; 19: 602–614. doi: 10.1016/j.cub.2009.05.046 PMID: 19640499
- Sures B. Environmental Parasitology: Interactions between parasites and pollutants in the aquatic environment. Parasite. 2008; 15: 434–438. PMID: <u>18814718</u>
- Pascoe D, Woodworth J. The effects of joint stress on sticklebacks. Zeits chriftfar Parasitenkunde. 1980; 62: 159–163.
- Stadnichenko AP, Ivanenko LD, Gorchenko IS, Grabinskaya OV, Osadchuk LA, Sergeichuk SA. The effect of different concentrations of nickel sulphate on the horn snail infected with the trematode *Cotylurus cornutus*. Parazitologiya (st. Petersburg). 1995; 29(2): 112–116.
- Marcogliese DJ, Brambilla LG, Gagné F, Gendron AD. Joint effects of parasitism and pollution on oxidative stress biomarkers in yellow perch *Perca flavescens*. Dis Aquat Organ. 2005; 63: 77–84. PMID: 15759803
- Gismondi E, Rigaud T, Beisel J-N, Cossu-Leguille C. Effect of Multiple Parasitic Infections on the Tolerance to Pollutant Contamination. PLoSONE. 2012; 7(7): e41950. doi: 10.1371/journal.pone.0041950
- Cézilly F, Favrat A, Perrot-Minnot M-J. Multidimensionality in parasite-induced phenotypic alterations: ultimate versus proximate aspects. J. Exp. Biol. 2013; 216: 27–35. doi: <u>10.1242/jeb.074005</u> PMID: <u>23225864</u>
- 9. Sures B, Siddall R, Taraschewski H. Parasites as Accumulation Indicators of Heavy Metal Pollution. Parasitology Today 1999; 15 (1): 16–21. PMID: <u>10234173</u>
- Schludermann C, Konecny R, Laimgruber S, Lewis JW, Schiemer F, Chovanec A, Sures B. Fish macroparasites as indicators of heavy metal pollution in river sites in Austria. Parasitology 2003; 126 (7) pp S61–S69.
- IPCC (2013) Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (Stocker T. F., Qin D., Plattner G.-K., Tignor M., Allen S. K., Boschung J., Nauels A., Xia Y., Bex V. & Midgley P. M. (eds.)). Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA. 2013.

- Noyes PD, McElwee MK, Miller HD et al. The toxicology of climate change: environmental contaminants in a warming world. Environment International 2009; (35: ): 971–986.
- Lafferty KD. Environmental parasitology; what can parasites tell us about human impact in the environment? Parasitology Today 1997; 13 (7): 251–255. PMID: 15275061
- 14. MacKenzie K. Parasites as pollution indicators in marine ecosystems: a proposed early warning system. Mar Pol Bull 1999; 38: 955–959.
- Okamura B, Hartikainen H, Schmidt-Posthaus H, Wahli T. Life cycle complexity, environmental change and the emerging status of salmonid proliferative kidney disease. Freshw Biol 2011; 56: 735–753.
- Heinonen J, Kukkonen JVK., Holopainen IJ. The effects of parasites and temperature on the accumulation of xenobiotics in a freshwater clam. Ecological Applications 1999; 9:475–481.
- Persoone G, Wells PG. Artemia in aquatic toxicology: a review, in: Sorgeloos, P. et al. (Ed.) (1987). Artemia research and its applications: 1. Morphology, genetics, strain characterization, toxicology. Proceedings of the Second International Symposium on the brine shrimp Artemia. pp. 259–275. 1987.
- Nunes BS, Carvalho FD, Guilhermino LM, Van Stappen G. Use of the genus Artemia in ecotoxicity testing. Environ Pollut. 2006; 144(2): 453–62. PMID: <u>16677747</u>
- Sorgeloos P, Persoone G, Baeza-Mesa M, Bossuyt E, Bruggeman E. The use of Artemia cysts in aquaculture: the concept of "hatching efficiency" and description of a new method for cyst processing. Proceedings of the annual meeting—World Mariculture Society. 1978; 9 (1–4): 715–721.
- Sánchez MI, Green AJ, Castellanos EM. Seasonal variation in the diet of the Redshank *Tringa totanus* in the Odiel Marshes, south-west Spain: a comparison of faecal and pellet analysis. Bird Study. 2005; 52:210–216.
- Sánchez MI, Green AJ, Castellanos EM. Spatial and temporal fluctuations in presence and use of chironomid prey by shorebirds in the Odiel saltpans, south-west Spain. Hydrobiologia. 2006; 567: 329– 340.
- Varo N, Green AJ, Sánchez MI, Ramo C, Gómez J, Amat J. Behavioural and population responses to changing availability of *Artemia* prey by moulting black-necked grebes, *Podiceps nigricollis*. Hydrobiologia. 2011; 664 (1): 163–171.
- Mohebbi F. The Brine Shrimp Artemia and hypersaline environments microalgal composition: a mutual interaction. Int J Aguat Sci. 2010; 1: 19–27.
- Sánchez MI, Varo N, Green AJ, Ramos C, Amat J. Cestodes change the isotopic signature of brine shrimp hosts: implications for aquatic food webs. Int J Parasitol. 2013; 43(1): 73–80. doi: <u>10.1016/j.</u> ijpara.2012.11.003 PMID: 23220358
- Sánchez MI, Georgiev BB, Green AJ. Avian cestodes affect the behaviour of their intermediate host Artemia parthenogenetica: an experimental study. Behavioural Processes. 2007; 74: 293–299. PMID: <u>17156940</u>
- Sánchez MI, Thomas F, Perrot-Minnot MJ, Bertrand-Michel J, Missé D. Neurological and physiological disorders in Artemia harboring manipulative cestodes. J Parasitol. 2009; 95: 20–24. doi: <u>10.1645/GE-1550.1</u> PMID: <u>18576844</u>
- Rode NO, Lievens EJP, Flaven E, Segard A, Jabbour-Zahab R, Sánchez MI, Thomas Lenormand T. Why join groups? Lessons from parasite-manipulated *Artemia*. Ecology Letters. 2013; 16(4): 493–501. doi: <u>10.1111/ele.12074</u> PMID: <u>23351125</u>
- Grande JA, Borrego J, Morales JA. A study of heavy metal pollution in the Tinto-Odiel estuary in Southwestern Spain using factor analysis. Environ Geol. 1999; 39: 1095–110.
- 29. Sánchez MI et al. High prevalence and abundance of cestode parasites throughout the annual cycle of Artemia salina and A. parthenogenetica in coastal Spain: relationship with abundance of avian final hosts. J Parasitol. 2013; 112: 1913–23.
- **30.** Georgiev BB. et al. Cestodes from *Artemia parthenogenetica* (Crustacea, Branchiopoda) in the Odiel Marshes, Spain: a systematic survey. Acta Parasitologica. 2005; 50: 105–117.
- Bhattacharya S, Gupta K, Debnath S, Ghosh UC, Chattopadhyay D, Mukhopadhyay A. Arsenic bioaccumulation in rice and edible plants and subsequent transmission through food chain in Bengal basin: a review of the perspectives for environmental health. Toxicol Environ Chem. 2012; 94: 429–41.
- Olías M, Nieto JM, Sarmiento AM, Cerón JC, Cánovas CR. Seasonal water quality wariations in a river affected by acid mine drainage: The Odiel river (south west Spain). Sci Total Environ. 2004; 333: 267– 281. PMID: <u>15364534</u>
- Sáinz A, Grande JA, de la Torre ML. Characterisation of heavy metal discharge into the Ria of Huelva. Environ Int. 2004; 30: 557–566. PMID: <u>15031016</u>

- Sarmiento AM, Nieto JM Casiot C, Elbaz Poulichet F, Egal M. Inorganic arsenic speciation at river basin scales: The Tinto and Odiel Rivers in the Iberian Pyrite Belt, SW Spain. Environ Pollut. 2009; 157: 1202–1209. doi: 10.1016/j.envpol.2008.12.002 PMID: 19135765
- Olías M, Cánovas CR, Nieto JM, Sarmiento AM. Evaluation of the dissolved contaminant load transported by the Tinto and Odiel rivers (Southwest of Spain). Applied Geochemistry. 2006; 21: 1733– 1749.
- Vicente-Martorell JJ, Galindo-Riaño MD, García-Vargas M, Granado-Castro MD. Bioavailability of heavy metals monitoring water, sediments and fish species from a polluted estuary. Journal of Hazardous Materials. 2009; 162(2): 823–836.
- Long ER, MacDonald DD, Smith SL, Calder FD. Incidence of adverse biological effects within ranges of chemical concentrations in marine and estuarine sediments. Environmental Management. 1995; 19 (1): 81–97.
- Canadian Council of Ministers of the Environment, Canadian Sediment Quality Guidelines for the Protection of Aquatic Life. 1999
- Sánchez M I, Georgiev BB, Nikolov PN, Vasileva GP, Green AJ. Red and transparent brine shrimps (Artemia parthenogenetica): comparative study of their cestode infections. Parasitology Research. 2006. 100 (1): 111–114. PMID: <u>16832701</u>
- American Public Health Association (APHA). Standard Methods for the Examination of Water and Wastewaters, 19th ed. American Public Health Association, Washington, DC. 1995.
- Reiss CS, McLaren I, Avendaño P. Utility of storage lipid volumes in inferring recent trophic history of copepods. Can. J. Fish. Aquat. Sci. 199; 56: 2444–2449.
- Lassiter RR, Hallam TG. Survival of the fattest: Implications for acute effects of lipophylic chemicals on aquatic populations. Environ Toxicol Chem. 1990; 9: 585–595.
- Brown AF, Pascoe D. Parasitism and host sensitivity to cadmium: an acanthocephalan infection of the freshwater amphipod Gammarus pulex. J. Appl. Ecol. 1989; 26: 473–487.
- Bustnes JO, Erikstad KE, Hanssen SA, Tveraa T, Folstad I, Skaare JU. Anti-parasite treatment removes negative effects of environmental pollutants on reproduction in an Arctic seabird. Proceedings of the Royal Society B-Biological Sciences. 2006; 273(1605): 3117–3122.
- Minguez L, Meyer A, Molloy DP, Giamberini L. Interactions between Parasitism and Biological Responses in Zebra Mussels (*Dreissena polymorpha*): Importance in Ecotoxicological Studies. Environmental Research. 2009; 109: 843–850. doi: 10.1016/j.envres.2009.07.012 PMID: 19691958
- Miller H. The relationship between catalase and haemoglobin in human blood. Biochem J. 1958 Feb; 68(2): 275–282. PMID: <u>13522614</u>
- Almagor M, Yatziv S, Kahane I. Inhibition of host cell catalase by *Mycoplasma pneumoniae*: a possible mechanism for cell injury. Infection and Immunity, 1983; 41 (1): 251–256. PMID: <u>6407999</u>
- Mohankumar K, Ramasamy P. White spot syndrome virus infection decreases the activity of antioxidant enzymes in *Fenneropenaeus indicus* Virus Research. 206; 115: 69–75. PMID: <u>16139913</u>
- 49. Frank SN, Faust S, Kalbe M, Trubiroha A, Kloas W, Sures B. Fish hepatic glutathione-S-transferase activity is affected by the cestode parasites *Schistocephalus solidus* and *Ligula intestinalis*: evidence from field and laboratory studies. Parasitology. 2011; 138 (7): 939–944. doi: <u>10.1017/</u>S003118201100045X PMID: 21518465
- Alonso-Alvarez C, Bertrand S, Devevey G, Prost J, Faivre B, Chastel O, Sorci G. An Experimental manipulation of life-history trajectories and resistance to oxidative stress. Evolution. 2006; 60(9): 1913–1924. PMID: <u>17089975</u>
- Thomas F, Poulin R, Brodeur J. Host manipulation by parasites: a multidimensional phenomenon. Oikos 2010; 119: 1217–1223.
- Amat F, Gozalbo A, Navarro JC, Hontoria F, Varó I. Some aspects of Artemia biology affected by cestode parasitism. Hydrobiologia. 1991; 212: 39–44.
- Phillips G R, Russo RC. Metal bioaccumulation in fishes and aquatic invertebrates: a literature review. By Phillips Glenn R. and Russo Rosemarie C., Fisheries Bioassay Laboratory, Montana State University. 1978.
- De Maagd PG, Van De Klundert ICM, Van Wezel AP, Opperhuizen A, Sijm DTHM. Lipid content and time-to-death dependent lethal body burdens of naphthalene and 1,2,4-trichlorobenzene in fathead minnow (*Pimephales promelas*). Ecotoxicol Environ Saf. 1997; 38: 232–237. PMID: <u>9469874</u>
- 55. Mommsen TP, Moon TW, Eds. Biochemistry and Molecular Biology of Fishes 6, Environmental Toxicology, 562 pp. Elsevier B. V., Amsterdam, Netherlands. 2005.

- 56. Heinonen J, Kukkonen JVK, Holopainen I.J. Temperature and parasite- induced changes in toxicity and lethal body burdens of pentachlorophenol in the freshwater clam *Pisidium amnicum*. Environ Toxicol Chem. 2001; 20(12): 2778–2784. PMID: <u>11764161</u>
- Martínez-Padilla J, Mougeot F, Pérez-Rodríguez L, Bortolotti GR. Nematode parasites reduce carotenoid-based signalling in male red grouse. Biology Letters. 2007; 3: 161–164. PMID: <u>17264052</u>
- Bačkor M, Vaczi P. Copper tolerance in the lichen photobiont *Trebouxia erici* (Clorophyta). Environ Exp Bot. 2001; 48:11–20.
- Surai A P, Surai P F, Steinberg W, Wakeman W G, Speake B K, Sparks N H C. Effect of canthaxanthin content of the maternal diet on the antioxidant system of the developing chick, British Poultry Science. 2003; 44 (4): 612–619. PMID: <u>14584852</u>
- Nakagawa K, Kang S.-D, Park D-K, Handelman GJ, Miyazawa T. Inhibition by b-carotene and astaxanthin of NADPH dependent microsomal phospholipid peroxidation. J. Nutr. Sci. Vitaminol. 1997. 43: 345–355. PMID: 9268922
- Sánchez M I, Hortas F, Figuerola J, Green A J. Sandpipers select red brine shrimps rich in carotenoids but also parasites. Ethology. 2009; 115: 196–200.
- Liaaen-Jensen S. Biosynthesis and Metabolism. In Carotenoids in Food Chain; Britton G., Liaaen-Jensen S., Pfander H., Eds.; Birkhäuser: Basel, Switzerland, 1998; 3:359–371.
- 63. Matsuno T. Aquatic animal carotenoids. Fish Sci. 2001; 67: 771-789.
- Arruda S F, Souza E M T, Siqueira E. Carotenoids from malanga (*Xanthosoma sagittifolium*) leaves protect cells against oxidative stress in rats. International Journal for Vitamin and Nutrition Research. 2005; 75(2): 161–8. doi: <u>10.1024/0300-9831.75.2.161</u> PMID: <u>15929638</u>
- 65. Hartley–Whitaker J, Ainsworth G, Meharg A. Copper- and arsenate-induced oxidative stress in *Holcus lanatus* L. clones with differential sensitivity. Plant Cell Environ. 2011; 24: 713–722.
- Huang C, Ke Q, Costa M, Shi X. Molecular mechanisms of arsenic carcinogenesis. Mo Cell Biochem. 2004; 25: 57–66.
- Mo J, Xia Y, Wade T, Schmitt M, Le XC, Dang R, Mumford JL. Chronic Arsenic Exposure and Oxidative Stress: OGG1 Expression and Arsenic Exposure, Nail Selenium, and Skin Hyperkeratosis in Inner Mongolia. Environ Health Perspect. 2006; 114(6): 835–841. PMID: <u>16759981</u>
- Flora SJ, Mittal M, Mehta A. Heavy metal induced oxidative stress and its possible reversal by chelation therapy. Indian J Med Res. 2008; 128(4): 501–23. PMID: <u>19106443</u>
- 69. Plaistow S J et al. The effect of the acanthocephalan parasite *Pomphorhynchus laevis* on the lipid and glycogen content of its intermediate host *Gammarus pulex*. Int J Parasitol. 2001; 31: 346–351. PMID: 11306112
- Jokela J, Uotila L, Taskinen J. Effect of the Castrating Trematode Parasite Rhipidocotyle fennica on Energy Allocation of Fresh-Water Clam Anodonta piscinalis. Functional Ecology. 1993; 7(3): 332–338.
- Baruš V, Tenora F, Kráčmar S, Prokeš M. Accumulation of heavy metals in plerocercoids of Ligula Intestinalis (Pseudophyllidea) of different age. Helminthologia. 2001; 38: 29–33.
- Baruš V, Prokeš M. Length and weight of *Ligula intestinalis* plerocercoids (Cestoda) parasitizing cyprinid fishes (Cyprinidae): a comparative analysis. Helminthologia. 2002; 39: 29–34.
- Caussy D. Case studies of the impact of understanding bioavailability: arsenic. Ecotoxicology and Environmental Safety. 2003; 56(1): 164–173. PMID: <u>12915149</u>
- Francesconi KA, Edmonds JS. Arsenic species in marine samples. Croatica Chemica Acta. 1998; 71 (2): 343–359.
- 75. Khan MA, Ahmed SA, Catalin B, Khodadoust A, Ajayi O, Vaughn M. Effect of temperature on heavy metal toxicity to juvenile crayfish, *Orconectes immunis* (Hagen). Environ Toxicol.2006; 21(5): 513–20. PMID: <u>16944513</u>
- Laetz CA, Baldwin DH, Hebert VR, Stark JD, Scholz NL. Elevated temperatures increase the toxicity of pesticide mixtures to juvenile coho salmon. Aquatic Toxicology. 2014; 146: 38–44. doi: <u>10.1016/j.</u> aquatox.2013.10.022 PMID: <u>24270668</u>
- 77. Bryant V, Newberry DM, McLusky DS, Campbell R. Effect of temperature and salinity on the toxicity of arsenic to three estuarine invertebrates (*Corophium volutator*, *Macoma balthica*, *Tubifex costatus*). Mar Ecol Prog Ser. 1985; 24: 129–137.
- **78.** Navarro JC, Ireland J, Tytler P. Effect of temperature on permeability and drinking rates of the metanauplii of the brine shrimp Artemia sp. Marine Biology. 1993; 116(2): 247–250.
- Blust R, Ginneken LV, Decleir W. Effect of temperature on the uptake of copper by the Brine shrimp Artemia franciscana. Aquatic toxicology. 1994; 30: 343–356.

- McGeachy SM, Dixon DG. The impact of temperature on the acute toxicity of arsenate and arsenite to rainbow trout (*Salmo gairdneri*). Ecotoxicology and environmental safety. 1989; 17(1): 86–93. PMID: <u>2714222</u>
- Hatcher MJ, Dick JTA, Dunn AM. Diverse effects of parasites in ecosystems: linking interdependent processes. Frontiers in Ecology and the Environment. 2012; 10: 186–194.
- De Wet LM, Schoonbee HJ, De Wet LPD, Wiid AJB. Bioaccumulation of metals by the southern mouthbrooder, *Pseudocrenilabrus philander* (Weber, 1897) from a mine polluted impoundment. Water SA 20:119–126. 1994.
- Canli M, Atli G. The relationships between heavy metal (Cd, Cr, Cu, Fe, Pb, Zn) levels and the size of six Mediterranean fish species. Environmental Pollution. 2003; 121: 129–136. PMID: <u>12475070</u>
- 84. Vasileva GP et al. Records of cysticercoids of *Fimbriarioides tadornae* Maksimova, 1976 and *Branchiopodataenia gvozdevi* (Maksimova, 1988) (Cyclophyllidea: Hymenolepididae) from brine shrimps at the Mediterranean coasts of Spain and France, with a key to cestodes from *Artemia* spp. from the Western Mediterranean. Acta Parasitologica. 2009; 54 (2): 143–150.
- Bush A O, Lafferty K D, Lotz J M, Shostak A W. Parasitology meets ecology on its own terms: Margolis et al. revisited. J Parasitol. 1997; 83:575–583. PMID: <u>9267395</u>
- Wurtsbaugh W A, Gliwicz Z M. Limnological control of brine shrimp population dynamics and cyst production in the Great Salt Lake, Utah. Hydrobiologia. 2001; 466: 119–132.
- The Bradford Method for protein quantification. Basic protein and peptide protocols. Volume 32 of the series Methods in Molecular Biology pp 9–15. PMID: <u>7951753</u>
- Cohen G, Dembiec D, Marcus J. Measurement of catalase activity in tissue extracts. Analytical Biochemistry. 1970. 34 (1): 30–38.
- McCord J M, Fridovich I. The utility of superoxide dismutase in studying free radical reactions I. radicals generated by the interaction of sulfite, dimethyl sulfoxide, and oxygen. 1969. Journal of Biological Chemistry, 243, 5753.
- Carmagnol F, Sinet P M, Jerome H. Seleniumdependent and non-selenium-dependent glutathione peroxidases in human tissue extracts. Biochim. biophys. Acta. 1983. 759: 49–57. PMID: 6882790
- Paglia D E, Valentine W N. Studies on the quantitative and qualitative characterization of erythrocyte glutathione peroxidase. The Journal of Laboratory and Clinical Medicine. 1967. 70 (1): 158–169. PMID: 6066618
- 92. Cribb A E, Leeder J S, Spielberg S P. Use of microplate reader in an assay of glutathione reductase using 5,5%-dithiobis(2-nitrobenzoic acid). Anal. Biochem. 1989. 183: 195–196. PMID: <u>2619044</u>
- 93. Buege J A, Aust S D. Methods Enzymol. 1978. 52: 302–310. PMID: 672633
- **94.** Hamilton MA, Russo RL, Thurston R V. Trimmed Spearman–Karber method for estimating median lethal concentrations. Environ Sci Tech. 1977; 11: 714–719.
- StatSoft, Inc. Electronic Statistics Textbook. Tulsa, Oklahoma. 2013. URL: <u>http://www.statsoft.com/</u> textbook.